Author’s response to reviews

Title: The Juice of Fresh Leaves of Catharanthus roseus Linn. Reduces Blood Glucose in Normal and Alloxan Diabetic Rabbits

Authors:

Dr Srinivas Nammi (nammi@rediffmail.com)
Dr Murthy K Boini (krishnamurthyboini@yahoo.com)
Srinivas D Lodagala (ldsrinivas@rediffmail.com)
Ravindra Babu S Behara (bsrbabu@hotmail.com)

Version: 3 Date: 22 Aug 2003

Answers for Reviewer's comments:

1. Since we have added one more column in tables, the readings at the time interval of 2 hours which considered to be not significant have been deleted to fit the page settings. The time that was included in text was due to typographic error and now it has been deleted.

2. Tables 1 and 2 are now standardized and presented without lines as suggested by the reviewer. We regret for the mistakes.